Author’s response to reviews

Title: Evaluation of Metabolic Syndrome in Adults of Talca City, Chile

Authors:

Veronica Mujica (vmujica@utalca.cl)
Elba Leiva (eleiva@utalca.cl)
Gloria Icaza (gicaza@utalca.cl)
Nora Diaz (ndiaz@utalca.cl)
Miguel Arredondo (marredon@inta.cl)
Rodrigo Moore-Carrasco (rmoore@utalca.cl)
Roxana Orrego (rorrego@utalca.cl)
Marcela Vasquez (mvasquez@utalca.cl)
Ivan Palomo (ipalomo@utalca.cl)

Version: 4 Date: 17 April 2008

Author’s response to reviews:

Santiago, April 18th, 2008.

Editor’s Office,

Dear Editor:

Enclosed you will find a revised version 2 of the manuscript entitle “Evaluation of Metabolic Syndrome in Adults of Talca City, Chile”. Verónica Mujica, Elba Leiva, Gloria Icaza, Nora Diaz, Miguel Arredondo, Rodrigo Moore-Carrasco, Roxana Orrego, Marcela Vásquez, Iván Palomo.

We included all the suggestions by the editorial productions.

Best regards,

Dr. Miguel Arredondo
Micronutrient Laboratory,
INTA, Universidad de Chile,
e-mail: marredon@inta.cl
Phone: 56-2-6781483
Fax: 56-2-2214030